Author’s response to reviews

Title: Optimising the value of the evidence generated in Implementation Science: the use of ontologies to address the challenges

Authors:

Susan Michie (s.michie@ucl.ac.uk)
Marie Johnston (m.johnston@abdn.ac.uk)

Version: 1 Date: 17 Oct 2017

Author’s response to reviews:

Susan Michie

University College London
Room 347, 1-19 Torrington Place,
London, WC1E 7HB.

Response to reviews of the Editorial “Optimising the value of the evidence generated in Implementation Science: the use of ontologies to address the challenges”

Dear Paul,

Thank you for sending the review decision for this Editorial article.

We have made all the required minor revisions, indicated with tracked changes.

We are finishing the process of reviewing the proofs for our related paper: “The Human Behaviour-Change Project: harnessing the power of artificial intelligence and machine learning for evidence synthesis and interpretation”. It would be very helpful if cross-referencing between the two papers could be added.

Please address correspondence concerning this manuscript to me at s.michie@ucl.ac.uk.

Thank you.
With best wishes,

Professor Susan Michie

University College London